Author's response to reviews

Title: Extracardiac Fontan with T-shape conduit in non-confluent pulmonary arteries

Authors:

Jeong Ryul Lee (jrl@plaza.snu.ac.kr)
Jae Gun Kwak (leri418@gmail.com)
Woo-Sung Jang (whiteuri@hanmail.net)
Eun Jung Bae (eunjung@plaza.snu.ac.kr)

Version: 3 Date: 26 January 2008

Author's response to reviews:

1. I removed the authors’ name, key words and running head from title page and also removed figures from text.
2. I reordered the reference citations.
3. I changed superscript marks to numbers and corrected the detail address.
4. I removed the authors’ qualification titles from the manuscript.
5. I used sentence case for all headings. ("Case Description" -> "Case description")